Dear Editor,

We have now revised the manuscript. The first table referred to in the previously submitted version of our paper was table 5 and you asked us to relabel this as table 1. The reference to this table is in the methods section. As it does not need a reference to a table here, we have taken out the table reference and put in the following explanatory text (see page 8):

We constructed a deprivation index from the child and household deprivation items available in the data set. A score of 1 ...

As a result, we do not need to reorder the tables. This means that the tables are in a logical order when people read the results section.

We hope you find this amendment satisfactory.

Best wishes,

Clare Blackburn